Author’s response to reviews

Title: Clinical prediction and diagnosis of neurosyphilis in HIV-negative patients: a case-control study

Authors:

Yong Lu (luyong91@yahoo.com)
Wujian Ke (kewujianstauch@163.com)
Ligang Yang (yanglg3@hotmail.com)
Zhenyu Wang (wangzhy39@mail2.sysu.edu.cn)
Ping Lv (lvping75@163.com)
Jing Gu (gujing5@mail.sysu.edu.cn)
Chun Hao (haochun@mail.sysu.edu.cn)
Jinghua Li (lijinghua3@mail.sysu.edu.cn)
Yumao Cai (caiyumao@hotmail.com)
Mei Gu (gmbjy@126.com)
Hongfang Liu (sunflower1977@126.com)
Wenjing Chen (wenjing1976@tom.com)
Xiaohui Zhang (108290423@qq.com)
Liuyuan Wang (454032866@qq.com)
Yahui Liu (409792901@qq.com)
Bin Yang (yangbin101@hotmail.com)
Huachun Zou (zouhuachun@mail.sysu.edu.cn)
Heping Zheng (290748650@qq.com)

Version: 3 Date: 16 Oct 2019

Author’s response to reviews:
1. In the “Funding” section of your declarations, please clarify the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: Thank you for your suggestion. We have added a description in the funding section, lines 351-352, page 18.

2. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Currently the individual contributions of Zhenyu Wang are not included. Guidance and criteria for authorship can be found here:

http://www.biomedcentral.com/submissions/editorial-policies#authorship

Response: Thank you for your suggestion. We have added Zhenyu Wang in the Authors’ contributions section, lines 356, page 19.

3. Please represent authors’ names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Response: Thank you for your suggestion. We have revised the Authors’ contributions section, lines 354-359, page 19 according to your requirement.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: Thank you for your suggestion. We have removed all the highlights in the final manuscript.